\newcommand{\etalchar}[1]{$^{#1}$}
\begin{thebibliography}{MAB{\etalchar{+}}94}

\bibitem[AD94]{AD94}
R.~Alur and D.L. Dill.
\newblock A theory of timed automata.
\newblock {\em Theor. Comp. Sci.}, 126:183--235, 1994.

\bibitem[AH89]{TPTL1}
R.~Alur and T.A. Henzinger.
\newblock A really temporal logic.
\newblock In {\em Proc. 30th IEEE Symp. Found. of Comp. Sci.}, pages 164--169,
  1989.

\bibitem[AH92]{AHsur}
R.~Alur and T.~Henzinger.
\newblock Logics and models of real time: A survey.
\newblock In J.W. de~Bakker, C.~Huizing, W.P. de~Roever, and G.~Rozenberg,
  editors, {\em Proceedings of the REX Workshop ``Real-Time: Theory in
  Practice''}, volume 600 of {\em Lect. Notes in Comp. Sci.}, pages 74--106.
  Springer-Verlag, 1992.

\bibitem[AL94]{AL91}
M.~Abadi and L.~Lamport.
\newblock An old-fashioned recipe for real time.
\newblock {\em ACM Trans. Prog. Lang. Sys.}, 16(5):1543--1571, 1994.

\bibitem[BAMP83]{BMP}
M.~Ben-Ari, Z.~Manna, and A.~Pnueli.
\newblock The temporal logic of branching time.
\newblock {\em Acta Informatica}, 20:207--226, 1983.

\bibitem[BH81]{BH81}
A.~Bernstein and P.~K. Harter.
\newblock Proving real time properties of programs with temporal logic.
\newblock In {\em Proceedings of the Eighth Symposium on Operating Systems
  Principles}, pages 1--11. ACM, 1981.

\bibitem[EC82]{EC}
E.A. Emerson and E.M. Clarke.
\newblock Using branching time temporal logic to synthesize synchronization
  skeletons.
\newblock {\em Sci. Comp. Prog.}, 2:241--266, 1982.

\bibitem[FG95]{FG95}
L.~Fix and O.~Grumberg.
\newblock Verificaiton of temporal properties.
\newblock to appear in {\sl Logic and Computation}, 1995.

\bibitem[Har87]{H}
D.~Harel.
\newblock Statecharts: {A} visual formalism for complex systems.
\newblock {\em Sci. Comp. Prog.}, 8:231--274, 1987.

\bibitem[Hen92]{IPL}
T.A. Henzinger.
\newblock Sooner is safer than later.
\newblock {\em Info. Proc. Lett.}, 43(3):135--142, 1992.

\bibitem[HMP94]{TPMTTS}
T.~Henzinger, Z.~Manna, and A.~Pnueli.
\newblock Temporal proof methodologies for timed transition systems.
\newblock {\em Inf. and Comp.}, 112(2):273--337, 1994.

\bibitem[KdR85]{KdR}
R.~Koymans and W.-P. de~Roever.
\newblock Examples of a real-time temporal logic specifications.
\newblock In B.D. Denvir, W.T. Harwood, M.I. Jackson, and M.J. Wray, editors,
  {\em The Analysis of Concurrent Systems}, volume 207 of {\em Lect. Notes in
  Comp. Sci.}, pages 231--252. Springer-Verlag, 1985.

\bibitem[Koy90]{Koy90}
R.~Koymans.
\newblock Specifying real-time properties with metric temporal logic.
\newblock {\em Real-time Systems}, 2(4):255--299, 1990.

\bibitem[KVdR83]{KVdR}
R.~Koymans, J.~Vytopyl, and W.-P. de~Roever.
\newblock Real-time programming and asynchronous message passing.
\newblock In {\em Proc. 2nd ACM Symp. Princ. of Dist. Comp.}, pages 187--197,
  1983.

\bibitem[Lam95]{possibility}
L.~Lamport.
\newblock Proving possibiity properties.
\newblock Technical Report 137, Digital Equipment Corporation, Systems Research
  Center, Palo Alto, July 1995.

\bibitem[MAB{\etalchar{+}}94]{STeP}
Z.~Manna, A.~Anuchitanukul, N.~Bj{\o}rner, A.~Browne, E.~Chang, M.~Col\'{o}n,
  L.~De Alfaro, H.~Devarajan, H.~Sipma, and T.~Uribe.
\newblock {STeP}: The {S}tanford {T}emporal {P}rover.
\newblock Technical Report STAN-CS-TR-94-1518, Dept. of Comp. Sci., Stanford
  University, Stanford, California, 1994.

\bibitem[MMP92]{MMP}
O.~Maler, Z.~Manna, and A.~Pnueli.
\newblock From timed to hybrid systems.
\newblock In J.W. de~Bakker, C.~Huizing, W.P. de~Roever, and G.~Rozenberg,
  editors, {\em Proceedings of the REX Workshop ``Real-Time: Theory in
  Practice''}, volume 600 of {\em Lect. Notes in Comp. Sci.}, pages 447--484.
  Springer-Verlag, 1992.

\bibitem[MP91a]{Comp91}
Z.~Manna and A.~Pnueli.
\newblock Completing the temporal picture.
\newblock {\em Theor. Comp. Sci.}, 83(1):97--130, 1991.

\bibitem[MP91b]{MPB}
Z.~Manna and A.~Pnueli.
\newblock {\em The Temporal Logic of Reactive and Concurrent Systems:
  Specification}.
\newblock Springer-Verlag, New York, 1991.

\bibitem[MP93]{ModR}
Z.~Manna and A.~Pnueli.
\newblock Models for reactivity.
\newblock {\em Acta Informatica}, 30:609--678, 1993.

\bibitem[MP94]{tacs94}
Z.~Manna and A.~Pnueli.
\newblock Temporal verification diagrams.
\newblock In T.~Ito and A.~R. Meyer, editors, {\em Theoretical Aspects of
  Computer Software}, volume 789 of {\em Lect. Notes in Comp. Sci.}, pages
  726--765. Springer-Verlag, 1994.

\bibitem[MP95]{MPB2}
Z.~Manna and A.~Pnueli.
\newblock {\em Temporal Verification of Reactive Systems: Safety}.
\newblock Springer-Verlag, New York, 1995.

\bibitem[MT90]{MT90}
F.~Moller and C.~Tofts.
\newblock A temporal calculus of communicating systems.
\newblock In J.C.M. Baeten and J.W. Klop, editors, {\em Proceedings of
  Concur'90}, volume 458 of {\em Lect. Notes in Comp. Sci.}, pages 401--415.
  Springer-Verlag, 1990.

\bibitem[NSY92]{NSY91}
X.~Nicollin, J.~Sifakis, and S.~Yovine.
\newblock From {ATP} to timed graphs and hybrid systems.
\newblock In J.W. de~Bakker, C.~Huizing, W.P. de~Roever, and G.~Rozenberg,
  editors, {\em Proceedings of the REX Workshop ``Real-Time: Theory in
  Practice''}, volume 600 of {\em Lect. Notes in Comp. Sci.}, pages 549--572.
  Springer-Verlag, 1992.

\bibitem[Ost90]{BOST}
J.S. Ostroff.
\newblock {\em Temporal Logic of Real-Time Systems}.
\newblock Advanced Software Development Series. Research Studies Press (John
  Wiley \& Sons), Taunton, England, 1990.

\bibitem[Pnu92]{vital}
A.~Pnueli.
\newblock How vital is liveness?
\newblock In W.R. Cleaveland, editor, {\em Proceedings of Concur'92}, volume
  630 of {\em Lect. Notes in Comp. Sci.}, pages 162--175. Springer-Verlag,
  1992.

\bibitem[SBM92]{SBM91}
F.~B. Schneider, B.~Bloom, and K.~Marzullo.
\newblock Putting time into proof outlines.
\newblock In J.W. de~Bakker, C.~Huizing, W.P. de~Roever, and G.~Rozenberg,
  editors, {\em Proceedings of the REX Workshop ``Real-Time: Theory in
  Practice''}, volume 600 of {\em Lect. Notes in Comp. Sci.}, pages 618--639.
  Springer-Verlag, 1992.

\bibitem[Sif91]{Si91}
J.~Sifakis.
\newblock An overview and synthesis on timed process algebra.
\newblock In K.G. Larsen and A.~Skou, editors, {\em 3rd Computer Aided
  Verification Workshop}, volume 575 of {\em Lect. Notes in Comp. Sci.}, pages
  376--398. Springer-Verlag, 1991.

\end{thebibliography}

\section{Introduction}

A formal framework for specifying and verifying temporal properties of
reactive systems often contains the following components:
\begin{itemize}
\item A \emm{computational model} defining the set of behaviors
(computations) that are to be associated with systems in the
considered model.
\item A \emm{requirement specification} language for specifying
properties of systems within the model. The languages we have
considered in our previous work are all variants of temporal logic
extended to deal with various aspects specific to the considered
model, such as real-time and continuously changing variables.
\item A \emm{system description} language for describing systems
within the model. We frequently use both a textual programming language and
appropriate extensions of the graphical language of statecharts
\cite{H} to present systems.
\item A set of \emm{proof rules} by which valid properties of systems
can be verified, showing that the systems satisfy their specifications.
\item A set of \emm{algorithmic methods} enabling a fully automatic
verification of decidable subclasses of the verification problem such
as the verification of finite-state systems (\emm{model checking}).
\end{itemize}

\nin
In \cite{ModR}, we considered a hierarchy of three models, each
extending its predecessor, as follows:
\begin{itemize}
\item A \emm{reactive systems} model that captures the
\emm{qualitative} (non-quantitative) temporal precedence
aspect of time. This model can only identify that one event precedes
another but not by how much.
\item A \emm{real-time systems} model that captures the \emm{metric}
aspect of time in a reactive system. This model can measure the time
elapsing between two events.
\item A \emm{hybrid systems} model that allows the inclusion of
\emm{continuous} components in a reactive real-time system. Such
continuous components may cause continuous change in the values of
some state variables according to some physical or control law.
\end{itemize}

\nin
The computational model proposed for reactive systems is that of a
\emm{fair transition system} (\fts) \cite{MPB}.

The approach to real time presented in \cite{ModR} and
\cite{TPMTTS} is based on the computational model of \emm{timed
transition systems} (\ttss) in which time itself is not explicitly
represented but is reflected in a time stamp affixed to each state in
a computation of a \ttss. As a result of time not being available as a
value of some variable, the requirement specification language must be
based on a necessary extension of temporal logic. In \cite{ModR}, we
present two such extensions: \emm{metric temporal logic} (\mtl) which
introduces bounded versions of the temporal operators, subscripted by
an interval specification, and \emm{temporal logic with age} (\tlg)
which introduces the operator $\Gamma$, measuring the length of recent
time in which a certain formula held continuously.

Consider, for example, the following important timed properties:
\begin{itemize}
\item \emm{Bounded response}:\quad Every $p$ should be followed by an
occurrence of a $q$, not later than $d$ time units.
\item \emm{Minimal separation}:\quad No $q$ can occur earlier than $d$
time units after an occurrence of $p$.
\end{itemize}

\nin
In \mtl, these two properties can be specified as follows:
\[\begin{array}{lll}
\bullet & \mbox{\rm Bounded response:}   & p\ \ent\ \F_{\leq d}\ q.\\[4pt]
\bullet & \mbox{\rm Minimal separation:}\quad & p\ \ent\ \G_{<d}\,\neg q.
\end{array}
\]

This approach to the specification of timing properties has been
advocated in \cite{KVdR}, \cite{KdR}, and \cite{Koy90}, although an
early proposal in \cite{BH81} can be viewed as a precursor to this
specification style.

\nin
Using \tlg, the same properties can be specified by:
\[\begin{array}{lll}
\bullet & \mbox{\rm Bounded response:}   & \G\Bigl[\Gamma\Bigl((\neg q) \since
(p\wedge\neg q)\Bigr)\ \leq\ d\Bigr].\\[4pt]
\bullet & \mbox{\rm Minimal separation:}\quad & q\quad \ent \quad
\Bigl(\opH(\neg p)\;\vee\;\Gamma(\neg p)\geq d\Bigr). 
\end{array}
\]
When constructing detailed proofs of such timed properties, we found
it necessary to form many auxiliary invariants involving the age
function $\Gamma$. Very soon, we realized that each age expression of
the form $\Gamma(p)$ represents a timer that started measuring time as
soon as $p$ becomes true and is reset (or shut off) whenever $p$
becomes false. This led to the suggestion that age expressions can be
eliminated in favor of the introduction of explicit timers.

Meanwhile, the model of timed automata \cite{AD94} has gained
dominance in the arena of real-time algorithmic verification. This
model uses timers which can only be reset by system transitions but
increase uniformly whenever time progresses. However, aiming at
algorithmic verification, this model can only capture finite-state
systems.

In this paper we present a new computational model for real-time
systems: \emm{clocked transition system} (\ctss). This model
represents time by a set of clocks (timers) which increase uniformly
whenever time progresses, but can be set to arbitrary values by system
(program) transitions.

It is easy and natural to stipulate that one of the clocks $T$ is never
reset. In this case, $T$ represents the \emm{master clock} measuring
real time from the beginning of the computation. This immediately
yields the possibility of specifying timing properties of systems by
unextended temporal logic, which may refer to any of the system
variables, including the master clock $T$. Thus, the two yardstick
properties considered above can be specified by the following
(unextended) temporal formulas:
\[\begin{array}{lll}
\bullet & \mbox{\rm Bounded response:}   & p\;\land\; (T=t_0)\quad \ent\quad
\F(q\;\land\; T\leq t_0+d).\\[4pt]
\bullet & \mbox{\rm Minimal separation:}\quad & p\;\land\;
(T=t_0)\quad \ent\quad \G(T<t_0+d\;\to\;\neg q). 
\end{array}
\]

\nin A first advantage of the new computational model over its
predecessors is that it leads to a more natural style of verification
proofs, stating invariants in terms of clocks, which are just another
kind of system variables, instead of formulating invariants in terms
of $\Gamma$-expressions. A second advantage is that we can use
temporal logic without any extensions as the specification language, and
can reuse many of the methods and tools developed for verifying
reactive systems, under the \fts\ computational model, for verifying
real-time systems under the \ctss\ model. Another advantage of the
\ctss\ model is that it can be viewed as a natural first-order
extension of the timed automata model \cite{AD94}.

In order to reuse \fts\ verification methods for verifying \ctss\
properties, we will show that for every \ctss\ $\Phi$ there exists a
corresponding \fts\ $P_{\Phi}$ such that the runs of $\Phi$ and the
runs of $P_{\Phi}$ coincide. It follows that any safety property is
valid over the \ctss\ $\Phi$ iff it is valid over the \fts\
$P_{\Phi}$. This reduction from real-time systems to simpler
transition systems enables us to use the deductive methodology
developed for proving safety properties of reactive systems, as well
as support systems for this methodology such as \cite{STeP}, for
establishing safety properties of timed systems.

It is interesting to note that the move from \ttss\ to \ctss\ brings
us closer to the approach proposed in \cite{AL91}, which also
recommends handling real time with a minimal extension of the
reactive-systems formalism.

The restriction to safety properties is not as constraining as one may
think. Many properties whose untimed version falls into the liveness
class become safety properties when we consider their timed version.
For example, termination of a program is a liveness property. However,
the property of termination within 15 time units is a safety property.
This indicates that, when we move to real-time systems, the set of
safety properties we may want to prove constitutes an even higher
portion of the total set of properties of interest than in the case of
untimed reactive-systems. We refer the reader to \cite{IPL} and
\cite{vital} for further discussions of this point.

We refer the reader to
\marginpar{\em refer to the two Henzinger's papers}
\cite{TPTL1}, \cite{BOST}, \cite{AL91}, and the
survey in \cite{AHsur}, for additional logics, models, and approaches
to the verification of real-time systems. In the process algebra
school, some of the representative approaches to real time are
\cite{NSY91}, \cite{MT90}, and many others are listed in \cite{Si91}.

The paper is organized as follows. In Section~\ref{cts}, we present
the real-time computational model of \emm{clocked transition systems}
(\ctss). In Section~\ref{programs}, we show how programs augmented
with timing bounds for the execution of statements can be represented
as clocked transition systems. In Section~\ref{safety}, we present
rules and verification diagrams for verifying safety properties of
\ctss's. In Section~\ref{response}, we present rules and verification
diagrams for establishing response properties of clocked transition
systems. Finally, in Section~\ref{nonzenos}, we present an approach to
verifying that a given \ctss\ is non-zeno.


